Neuro-Oncology Advances

7(1), vdaf105, 2025 | https://doi.org/10.1093/noajnl/vdaf105 | Advance Access date 5 June 2025

Long-term psychological and neurological outcomes among people with a history of non-malignant meningioma in the UK Biobank cohort

Diana R. Withrow^o, Fahad S. Al-Huda, Helen Bulbeck^o, Krishnan Bhaskaran^o, Robin Grant^o, Usama Ali^o, Pieter Pretorius, Jason L. Oke, and Brian D Nicholson

All author affiliations are listed at the end of the article

Corresponding Author: Diana R. Withrow, Faculty of Public Health and Policy, London School of Hygiene and Tropical Medicine, Keppel St, London WC1E 7HT (diana.withrow@lshtm.ac.uk).

Abstract

Background. Meningiomas are among the most common brain tumours in the United Kingdom and incidence is increasing. Up to 90% of meningiomas are Grade 1 (non-malignant) and have high survival. Brain tumour survivors are at risk of psychological and neurological late effects but risks in persons with non-malignant meningioma are not well characterized.

Methods. We used UK Biobank, a cohort of approximately 500 000 adults recruited ages 40-69 during 2006-2010. Non-malignant meningioma patients were identified through linked cancer registry data. Follow-up for 10 outcomes was based on linkage to Hospital Episode Statistics. Standardized incidence ratios (SIRs) compared risks in meningioma patients to rates from UK Biobank overall, adjusted for age, sex, and calendar time.

Results. Four hundred and sixty-seven individuals were diagnosed with non-malignant meningioma after joining UK Biobank (77% female). Median age at diagnosis was 65 and median follow-up was 5 years. Persons with meningioma had significantly increased risks of 9 of 10 sequelae studied. The lowest SIR was for stroke (1.3, 95% confidence interval [CI] 0.6-2.9) and the highest SIRs were for epilepsy (18.9, 95%CI: 13.3-26.9) and visual disturbances (6.6, 95%CI: 3.3-13.1). SIRs for depression, anxiety, headache, fatigue, hearing loss, limb weakness, and cognitive issues (in ascending order) ranged from 2.3 (95%CI: 1.6-3.4) to 4.4 (95%CI: 2.8-6.9).

Conclusions. By providing preliminary evidence of excess risk of a range of long-term sequelae, this research can provide insight into future risks and validate survivor experience for people diagnosed with non-malignant meningioma and build momentum for future research using larger population-based databases and primary care records.

Key Points

- Meningiomas are common brain tumours with rising incidence rates.
- Electronic medical record linkage in a large cohort study examined long-term outcomes.
- Results show increased risks of psychological and neurological sequelae in survivors.

Meningiomas are tumours that occur in the tissue covering the brain and spinal cord (meninges) and are the most frequently diagnosed brain tumours among women, and the second most common brain tumour among men in the United Kingdom.¹ Grade 1 (non-malignant) meningiomas are slow growing, localized, and can often be treated with observation or surgery alone, whereas Grade 2 (atypical) or 3 (malignant) meningiomas can spread to surrounding tissues, often requiring more aggressive treatment. Up to 90% of meningiomas are Grade 1.² The incidence of Grade 1 meningiomas has been increasing in England by 8% per year on average since the early 1990s.³ Increases have also been

This is an Open Access article distributed under the terms of the Creative Commons Attribution License (https:// creativecommons.org/licenses/by/4.0/), which permits unrestricted reuse, distribution, and reproduction in any medium, provided the original work is properly cited.

[©] The Author(s) 2025. Published by Oxford University Press, the Society for Neuro-Oncology and the European Association of Neuro-Oncology.

Importance of the Study

Meningiomas are among the most common brain tumours, with rising incidence rates. The long-term effects of non-malignant meningioma have not been thoroughly described. This study is the first to examine long-term neurological and psychological outcomes in individuals with non-malignant meningioma using the UK Biobank. By using linked electronic health records,

recorded in the United States⁴ likely owing at least in part to advancements in diagnostic techniques which have increased detection rates.^{1,5}

These Grade 1 meningiomas (henceforth referred to as non-malignant meningiomas [NMMs]) are not classified as cancer but depending on their location and size, they may cause serious symptoms. For symptomatic and/or growing NMMs, the primary treatment is surgery with the goal of relieving symptoms. Stereotactic radiosurgery and fractionated radiotherapy can be indicated in some cases but are most often reserved for higher grade (WHO grade 2 or 3) tumours.⁶ In a US study of nearly 50 000 meningioma patients, of whom 94% had non-malignant histology, over 40% underwent surgery, a likely underestimate of the true proportion of patients undergoing surgery as the study was registry-based.⁷The long-term consequences of NMM are not well understood as people with NMM have seldom been studied from a long-term survivorship perspective despite the likely neurological and psychological risks associated with the surgery and the tumour itself.

Much of the research to date has focussed on brain tumours broadly,⁸⁻¹² or, when studying NMM, have had short-term follow-up of around 1 year after diagnosis.^{13,14} A few studies have focussed specifically on people with a history of NMM and have had longer follow-up. One Swedish study of 190 patients with median follow-up of 9 years reported that 49% of patients had impaired health-related quality of life (HRQOL), and 43% suffered from neurocognitive deficits.¹⁵ In a study carried out in Melbourne, 291 patients followed up for up to 10 years selfreported decreased HRQOL across a number of domains compared to a healthy reference.¹⁶ In a study carried out in the Netherlands, among 89 patients with 11.5 years of median follow-up, 67% showed at least one neurological symptom.^{17,18} These studies have been small, have used self-report from the patient or provider, and have relied on composite outcomes such as HRQOL, and/or reported only absolute rather than relative measures of the outcomes, which can hinder interpretation when study populations vary demographically from each other and from the general population. As far as we are aware, there are no multi-institutional studies of the relative risks of multiple, specific long-term sequelae among persons with a history of NMM.

We hypothesized that meningioma survivors have a greater incidence of neurological and psychological morbidity. To test this, we estimated standardized incidence ratios for neurological and psychological symptoms in a the study provides the largest analysis of its kind. The findings confirm increased risks for several health issues, including epilepsy and visual disturbances, validating patient concerns. This study underscores the need for more research to enhance the understanding and management of non-malignant meningioma in the years after a diagnosis.

large population with NMM, comparing to a populationbased cohort. In doing so, we explore the feasibility of using electronic health records to study these outcomes in larger populations than previously studied. The specific outcomes selected for analysis were based on a review of the literature on long-term sequelae among survivors of CNS tumours of any type. The desired outcome for the research is to inform patients and providers of specific challenges patients with a history of NMM might face and to identify priorities for future research in this area.

Methods

Study Design and Data Sources

This was a prospective cohort study using data from the UK Biobank cohort. The UK Biobank is a population-based cohort of approximately half a million people recruited in the United Kingdom between 2006 and 2010. All participants have linked data from hospital records, mortality records, and the cancer registry, and a subset have been linked to primary care data. Specific data sources for each of the participating nations (England, Scotland, and Wales) are provided in SupplementaryTable 1.

We identified participants with a new diagnosis of NMM in the cancer registry. Cancer registries collect information on all primary cancers (and some non-malignant neoplasms, such as NMM) diagnosed in individuals primarily resident in their catchment areas. NMMs are usually differentiated from malignant meningiomas based on WHO Grade. As the cancer registry data does not routinely collect WHO Grade, we used morphology/histology and site/topography to identify diagnoses of NMM. NMM diagnoses were identified based on the following codes: ICD-10 - D320: Benign neoplasm of cerebral meninges; D321: Benign neoplasm of spinal meninges; D329: benign neoplasm of meninges, unspecified; ICD-O-3 - 9530/0: Meningioma, NOS; 9531/0; Meningothelial meningioma; 9532/0: Fibrous meningioma; 9533/0: Psammomatous meningioma; 9534/0: Angiomatous meningioma; 9537/0: Transitional meningioma. These may have been radiographically or histologically confirmed cases. We refer to these tumours as non-malignant rather than benign to reflect preferences in the patient and advocacy communities.¹⁹

We restricted to NMM diagnoses occurring after entry into UK Biobank to reduce the selection bias that could occur if people with pre-existing NMM diagnoses declined to participate in the UK Biobank study. In a supplemental analysis, we extended the "exposed" definition to include people who had been diagnosed with a NMM prior to registration with UK Biobank.

Ten neurological/psychological outcomes were selected for analysis based on findings of a literature review, which suggested increased risks of these among survivors of central nervous system tumours. These were: cognideficits/impairment,^{8–12,14–18,20–22} epilepsy,^{8,14,17,18,23} tive visual impairment or disturbances,8,9,17,21,23,24 depression,^{10,11,14,20,22} anxiety,^{10,11,14,20,24} fatigue,^{9,14,16,24} limb weakness,^{8,9,24} hearing loss,^{9,21,23} headache,^{17,23} and stroke.^{17,25} Insomnia^{14,16} and anosmia²³ were also considered for inclusion but ultimately excluded due to low levels of recording in the electronic health records.

We used outcomes identified using inpatient records from hospital episode statistics. ICD-10 code lists were developed using opencodelists.org and manual searching and are provided in the supplement (Supplementary Table 2). Each hospital encounter can have up to 20 codes so the code did not have to be present as a primary complaint. In our primary analysis, we considered any instance of the code occurring during follow-up. In other words, the outcome did not have to be the first ever instance for the event to be included, since many of the outcomes we were exploring may also occur as symptoms prior to a meningioma diagnosis. In sensitivity analyses, we restricted to the first ever diagnosis or occurrence of the code.

For the cohort at large, follow-up began at the date of assessment for UK Biobank. Participants were excluded if their age at UK Biobank assessment was less than 40 since being aged 40 to 65 was a criterion for UK Biobank entry (n = 7). One participant was excluded because their record was missing information on age at assessment and year of birth.

Patients were censored if they died, left the study, or at the end of the study period. At the time of data extract, the end date of the linked data sources varied through 2021. We set the end of follow-up to be March 2, 2021, the latest date an incident NMM was recorded.

For the "exposed" group of NMM patients, follow-up began 6 months after the date of diagnosis. The 6-month latency period was selected in order to reduce the likelihood of including acute/short-term side effects of treatment. Sensitivity analyses explored the impact of eliminating the latency period.

In exploratory analyses we stratified the results by receipt of surgery. A list of OPCS-4 procedure codes designating a meningioma surgery were selected and is provided in the supplement (Supplementary Table 2). We considered the surgery related to the diagnosis if it occurred at most 2 months prior and up to 6 months after diagnosis. Seventy-five percent of all surgeries matching the OPCS-4 procedure codes occurring among meningioma patients occurred within that time window.

Statistical Analysis

We estimated standardized incidence ratios of each outcome. A standardized incidence ratio is the ratio of the expected rate of events to the observed rate. Expected rates were generated for the UK Biobank as a whole by age (10-year age groups), sex, and calendar period (5 years). Because meningioma cases make up only 0.1% of the cohort, their inclusion in the background rates is considered unlikely to impact the results. Standardized incidence ratios (SIRs) were stratified by age (at the time of follow-up/ event) and sex. We reported SIRs for the subgroup of patients who were recorded as having surgery, but since the number of events in patients without surgery was consistently small, and because the definition of surgery within 6 months may have misclassified some people who received surgery later, we have not reported SIRs for persons who were not recorded to have had surgery.

Patient and Public Involvement

Patients were consulted during 2 rounds of application for funding. At both instances, patients emphasized the usefulness of the work and provided some wording suggestions for the lay abstract. While the analyses were in progress, D.W. presented at 2 meetings of the brainstrust Meningioma support group. Patients expressed their gratitude for the research, and relayed experiences of feeling disappointed and frustrated by health care providers who attributed their symptoms to ageing or menopause. Patients also suggested other symptoms to include and/ or explore however these were not coded specifically or frequently enough to be included in this study. Following these group meetings, H.B. was invited to be included as a co-author to represent the patient/carer perspective.

Ethics

UK Biobank has approval from the North West Multi-centre Research Ethics Committee (MREC) as a Research Tissue Bank (RTB) approval. This approval means that researchers do not require separate ethical clearance and can operate under the RTB approval.

Results

The cohort at large contained just over half a million people (n = 501 937) with a mean age at entry to the UK Biobank cohort of 56.5 years (standard deviation [SD]: 8.1) and 11.8 years of follow-up (SD: 1.7, Table 1). Four hundred and sixty-seven NMMs were diagnosed during follow-up (0.1%) of the cohort). Persons with NMM joined the UK Biobank at an older age (58.1 years, SD: 7.6) and had less follow-up (5.1 years, SD: 3.0), since follow-up began at the time of their diagnosis. The mean age at NMM diagnosis was 65 (SD: 7.9) years.

Patients with a history of NMM were more likely to be female than the cohort as a whole (77% vs. 54%, Table 1). Those who would be diagnosed with NMM were not notably different from the cohort with respect to ethnicity (White vs. other, mixed, or unknown), deprivation quintile, or body mass index at enrolment to the UK Biobank. NMM patients were less likely to smoke at baseline than the <u>Neuro-Oncology</u>

Table 1. Demographics at UK Biobank Baseline

	No meningioma	Meningioma	
	N = 501 937	N = 467	
	Mean (SD)	Mean (SD)	
Age at assessment	56.53 (8.10)	58.11 (7.59)	
Age at meningioma diagnosis		65.38 (7.89)	
Years of follow-up	11.76 (1.72)	5.09 (3.03)	
	n (%)	n (%)	
Sex			
Female	272 965 (54.4%)	359 (76.9%)	
Male	228 972 (45.6%)	108 (23.1%)	
Country at assessment			
England	444 515 (88.6%)	415 (88.9%)	
Wales	20 787 (4.1%)	17 (3.6%)	
Scotland	36 583 (7.3%)	35 (7.5%)	
Ethnicity			
Other/Mixed/Unknown	30 249 (6.0%)	26 (5.6%)	
Any white background	471 688 (94.0%)	441 (94.4%)	
Townsend Quintile			
5 - Least deprived	229 159 (45.7%)	227 (48.6%)	
4	110 739 (22.1%)	100 (21.4%)	
3	73 430 (14.6%)	68 (14.6%)	
2	60 146 (12.0%)	51 (10.9%)	
1 - Most deprived	27 837 (5.6%)	21 (4.5%)	
Smoking status			
Never	273 209 (54.4%)	262 (56.1%)	
Previous	172 851 (34.4%)	172 (36.8%)	
Current	52 932 (10.5%)	29 (6.2%)	
Missing/unknown	2 945 (0.6%)	4 (0.9%)	
Body mass index (BMI)			
Underweight	2 626 (0.5%)	0 (0.0%)	
Healthy	162 219 (32.3%)	149 (31.9%)	
Overweight/Obese	333 988 (66.5%)	315 (67.5%)	
Missing	3 104 (0.6%)	3 (0.6%)	

cohort as a whole. This was the case even after accounting for sex (8.9% among cohort vs. 5% among NMM in females and 12.5% vs. 10.2% in males).

As shown in Figure 1 and Table 2, NMM patients were significantly more likely to experience 9 out of 10 of the neurologic outcomes under study. The SIR was highest for epilepsy, where NMM patients had incidence 18.9 times higher than the cohort overall (95% confidence interval [CI]: 13.3-26.9). The lowest SIR and the only one that was non-significant was for stroke (1.3, 95% CI: 0.6-2.9). All other SIRs ranged between 2.3 (depression, 95% CI: 1.6-3.4) and 4.4 (cognitive issues, 95% CI: 2.8-6.9).

Stratified analyses are presented in Table 2. Stratified SIRs were not presented when based on fewer than 5 events, so have been omitted for several outcome/subgroups. When reported, SIRs tended to be similar in the subgroup undergoing surgery, tended to be higher in those aged less than 70 than those aged older than 70, and tended to be higher in males than in females.

As described in the Methods section, we conducted 2 sensitivity analyses: (1) restricting to first ever occurrence of an outcome code as an event, (2) removing the latency period. We also did a supplemental analysis including prevalent (existing) and incident (new) meningioma diagnoses. Overall, after considering uncertainties, the results from all 4 analyses are very similar (Supplementary Table 3). As might be expected, SIRs tended to be marginally higher when the latency period was removed and marginally lower when prevalent diagnoses (those diagnosed prior to cohort entry) were included. When the latency period for stroke was removed, the SIR increased and was statistically significant (SIR: 2.4, 95%CI: 1.4-4.2).



Neuro-Oncology Advances

Discussion

This is the first study that we are aware of to examine longterm outcomes among individuals diagnosed with NMM using linkage to electronic health records. This approach allowed us to efficiently follow over 450 patients for a median of 5 years. We observed significantly increased risk of 9 out of 10 conditions examined, with these conditions being between twice and 20 times more common among persons with NMM than background rates in the cohort. This evidence can serve to validate the lived experience of NMM patients and improve patient communication about the outcomes of treatment for NMM. Furthermore, this study serves as a proof-of-concept and evidence that further research, using larger datasets that include primary care records, which would capture a broader range of sequelae with more sensitivity than hospital records, would be valuable.

As most of the studies to date have been survey-based, have included survivors of other types of brain tumours, and/or have not had estimates of relative risk, it is difficult to compare our findings quantitatively to existing published work. We chose our list of outcomes based on the literature, so insofar as we saw an increased risk of these factors in our cohort, our findings are consistent.^{8–10,12,13} The standardized incidence ratio was highest for seizures/ epilepsy, which may be at least in part attributable to the low background rate in the population relative to the other

outcomes examined. Stroke was the only outcome that was not significantly elevated in NMM patients, despite being suggested by our literature review as a potential risk. This may be because stroke is likely to occur peri-operatively. When we removed the 6-month latency period in the sensitivity analyses, there was a significantly increased risk of stroke among people with a history of NMM.

The study has 4 primary strengths. First, using incident cases of NMM from within an existing population-based cohort reduced the risk of selection bias, whereas enrollees to a cohort study with a prior history of NMM may not be representative persons with a diagnosis of NMM overall. Second, using administrative health records lends an objectivity to the ascertainment of outcomes. Cognitive decline experienced by patients, for example, may reduce the accuracy of self-reporting of outcomes. Third, the use of administrative health records allowed us to conduct the largest study of specific long-term outcomes in NMM of which we are aware. Fourth, this approach allowed us to achieve a median follow-up of 5 years among cases relatively efficiently, with very high follow-up and low rate of attrition. Finally, because SIRs compare observed to expected rates, they also account for differences in follow-up time between the two comparison groups.

There are, however, also several limitations to this analysis. We have tried to address some of these using the sensitivity analyses described above. One major limitation is that we used hospital, rather than primary care records. In the initial design of the study, we anticipated more primary

				, 0	
	n	Observed	Expected	SIR	95% CI
Epilepsy					
Overall	467	31	1.6	18.9	(13.3, 26.9)
With surgery	333	29	1.2	24.7	(17.2, 35.6)
<70 years of age ^a		19	0.8	25.2	(16.1, 39.5)
≥70 years of age		12	0.9	13.5	(7.7, 23.8)
Female	363	19	1.2	15.4	(9.8, 24.1)
Male	115	12	0.4	29.6	(16.8, 52.2)
Limb weakness					
Overall	467	23	5.7	4.0	(2.7, 6.1)
With surgery	333	16	4.0	4.0	(2.5, 6.6)
<70 years of age ^a		9	1.5	5.9	(3.1, 11.4)
≥70 years of age		14	4.2	3.4	(2.0, 5.7)
Female	363	14	4.3	3.3	(1.9, 5.5)
Male	115	9	1.4	6.3	(3.3, 12.1)
Depression					
Overall	467	25	10.7	2.3	(1.6, 3.4)
With surgery	333	17	7.9	2.2	(1.3, 3.5)
<70 years of age ^a		16	5.5	2.9	(1.8, 4.7)
≥70 years of age		9	5.2	1.7	(0.9, 3.3)
Female	363	18	9.1	2.0	(1.3, 3.2)
Male	115	7	1.7	4.2	(2.0, 8.8)
Anxiety					
Overall	467	24	8.8	2.7	(1.8, 4.1)
With surgery	333	20	6.2	3.2	(2.1, 5.0)
<70 years of age ^a		14	3.8	3.7	(2.2, 6.2)
≥70 years of age		10	5.0	2.0	(1.1, 3.7)
Female	363	19	7.5	2.5	(1.6, 3.9)
Male ^b					(,,
Cognitive issues					
Overall	467	19	4.3	4.4	(2.8, 6.9)
With surgery	333	12	3.0	4.0	(2.3, 70)
<70 years of age ^a	000	7	11	6.4	(3.0, 13.4)
>70 years of age		12	3.2	3.7	(2.1.6.6)
Female	363	15	3.0	2.0	(30.82)
Maleb	000	10	0.0	210	(0.0, 0.2)
Headache					
Overall	467	13	13	3.0	(18 5 2)
With surgery	333	10	3.1	3.2	(1.3, 5.2)
<70 years of ago	333	8	2.1	3.2	(1.7, 0.0)
>70 years of age		0	2.1	5.5	(1.3, 7.7)
≥70 years of age	262	10	2.6	2.4	(10 5 0)
Malab	303	12	3.0	3.4	(1.9, 5.9)
Overall	407	15	4 5	2.2	
With ourses	407	15	4.0	3.3	(2.0, 5.5)
with surgery	333	Э	3.1	2.9	(1.5, 5.5)
<td></td> <td>10</td> <td>2.4</td> <td>0.0</td> <td></td>		10	2.4	0.0	
≥/U years of age		12	3.4	3.6	(2.0, 6.3)

Table 2. Continued					
	п	Observed	Expected	SIR	95% Cl
Female	363	10	3.4	2.9	(1.6, 5.5)
Male ^b					
Hearing loss					
Overall	467	17	5.0	3.4	(2.1, 5.5)
With surgery	333	13	3.5	3.7	(2.2, 6.4)
<70 years of age ^a		10	1.4	7.2	(3.9, 13.3)
≥70 years of age		7	3.6	2.0	(0.9, 4.1)
Female	363	16	3.5	4.6	(2.8, 7.4)
Male ^b					
Stroke					
Overall	467	6	4.7	1.3	(0.6, 2.9)
With surgery ^b					
<70 years of ageª					
≥70 years of age					
Female ^b					
Male ^b					
Visual disturbance					
Overall	467	8	1.2	6.6	(3.3, 13.1)
With surgery	333	6	0.9	7.0	(3.1, 15.5)
<70 years of age ^a					
≥70 years of age					
Female	363	8	0.9	9.0	(4.5, 17.9)
Male ^b					

^aNumbers of individuals eligible are not provided for subcategories since patients move between categories of age during follow-up. ^bResults omitted when based on ≤5 observed events.

care coverage and/or an update of the UK Biobank primary care dataset during the project. At the time of data release, however, primary care linkage was available for approximately 45% of the cohort, with data spanning up until 2016 or 2017. Restricting to those who had primary care data reduced the number of NMM patients by two-thirds, leading to a loss in power that outweighed advantages in sensitivity provided by the primary care records. We also explored using follow-up surveys in the UK Biobank for self-reported follow-up but the number and timing of responses was insufficient to provide meaningful additional data. In the present study, diagnosis codes in hospital records alone were used to assess all outcomes/sequelae.

Some of our outcomes would not merit admission to hospital. While this is true for both NMM cases and comparators, because the event needs to be recorded in the hospital record, increased risks may simply reflect greater contacts with the secondary care system among NMM cases. Due to the reliance on hospital records, the count of events is presumed to highly underestimate the true incidence of any given condition. The extent to which the hospital recording of a diagnosis or symptom represents the true occurrence will depend on the importance of the diagnosis to treatment and/or hospital processes and reimbursement. As a result, a stroke will be more likely to be recorded than a visual disturbance, for example. To address this in part, we have focussed on ratios and intentionally not reported rates. Nevertheless, we acknowledge this as a major limitation and call for studies using primary care data in larger populations.

Furthermore, UK Biobank participants are healthier, more affluent, and more likely to be white than members of the UK population at large.²⁶ While background rates of morbidity will be underestimated in the UK Biobank cohort relative to the general population, we believe the SIRs could be internally valid since people diagnosed with NMM after entry into the UK Biobank cohort will have been subject to the same "healthy volunteer bias." The cancer data linked to the UK Biobank cohort did not include information on tumour location, method of confirmation (eg radiological or histological), recurrence, or aspects of treatment such as radiotherapy and/or whether the resection was complete or partial. Studies with this information could generate more informative, stratified risk estimates. Finally, while this study constitutes one of the largest cohorts of NMM patients followed longitudinally, the use of SIRs rather than causal modelling and relative risks reflects the descriptive nature of the research and the small sample

euro-Oncology

dvances

size for certain outcomes (ie insufficient power to control for numerous covariates).

Despite the significant limitations to this study, our results complement survey data reporting deficits in quality of life and decreased participation in work, social, and recreational activities among persons with a history of NMM.¹⁵ Together, they suggest that these patients would benefit from care that reflects their increased risks of psychological and neurological late effects. In a recent study of unmet needs in people with a diagnosis of NMM, information on possible long-term complications, expectation management, and research on outcomes were all identified as high importance.²⁵ In combination with evidence from other sources, our findings could be integrated into materials for patients, including survivorship care plans and into clinical conversations with patients, before, during and after treatment.

Supplementary Material

Supplementary material is available online at *Neuro-Oncology Advances* (https://academic.oup.com/noa).

Keywords

brain tumours | meningioma | late effects | survivorship

Funding

This study was supported by the National Institute for Health and Care Research (NIHR) School for Primary Care Research (project reference 662); Pump-priming funds were provided by the Oxford Cancer Development Fund (February 2022); National Institute of Health Research Academic Clinical Lectureship and a CRUK Research Careers Committee Postdoctoral Fellowship (RCCPDF\100005 to B.D.N.); and Wellcome Senior Research Fellowship (220283/Z/20/Z to K.B.).

Acknowledgements

The authors would like to thank Dr Zahra Raisi-Estabragh and her team at the Centre for Advanced Cardiovascular Imaging at Queen Mary University London. They have expertise in conducting survivorship research using the UK Biobank and generously provided early feedback on this work. Ms Jane Halliday provided valuable input on the surgical codes to be included. We would like to thank Professor Rafael Perera-Salazar for his input on the funding applications. We would also like to acknowledge the patients who participated in the *brainstrust* webinars, read lay abstracts, and generally provided invaluable motivation and feedback. This research has been conducted using the UK Biobank Resource under Application Number 89047. This work uses data provided by patients and collected by the NHS as part of their care and support. Copyright © 2024, NHS England. Re-used with the permission of the UK Biobank. All rights reserved. This research used data assets made available by Scottish National Safe Haven as part of the Data and Connectivity National Core Study, led by Health Data Research UK in partnership with the Office for National Statistics and funded by UK Research and Innovation (research which commenced between 1st October 2020–31st March 2021 grant ref MC_PC_20029; 1st April 2021 -30th September 2022 grant ref MC_PC_20058).

Conflict of interest statement. The authors have no conflicts of interest to declare.

Author Contributions

Conception and design of the work: D.R.W., F.S.A., P.P., J.L.O., B.D.N.

Acquisition, analysis, or interpretation of data for the work: D.R.W., F.S.A., H.B., K.B., R.G., UA, P.P., J.L.O., B.D.N.

Drafting the work or reviewing it critically for important intellectual content: D.R.W., F.S.A., H.B., K.B., R.G., U.A., P.P., J.L.O., B.D.N.

UK Biobank data access process (see http://www.ukbiobank. ac.uk/register-apply/).

Data Availability

The datasets generated by UK Biobank analysed during the current study are available via the

Affiliations

Clinical Effectiveness Unit, Faculty of Public Health and Policy, London School of Hygiene and Tropical Medicine, London (D.R.W.); Clinical Effectiveness Unit, Royal College of Surgeons of England, London (D.R.W.); Nuffield Department of Primary Care Health Sciences, University of Oxford, Oxford (D.R.W., F.S.A.-H., B.D.N.); brainstrust, Cowes, Isle of Wight (H.B.); Department of Non-Communicable Disease Epidemiology, Faculty of Epidemiology and Population Health, London School of Hygiene and Tropical Medicine, London (K.B.); Department of Clinical Neurosciences, Royal Infirmary of Edinburgh, Edinburgh (R.G.); Nuffield Department of Population Health, University of Oxford, Oxford (U.A.); Department of Neuroradiology, John Radcliffe Hospital, Oxford University Hospitals NHS Foundation Trust (P.P.); Abbott Diabetes Care, Witney, Oxfordshire (J.L.O.)

References

 Wanis HA, Moller H, Ashkan K, Davies EA. The incidence of major subtypes of primary brain tumors in adults in England 1995-2017. *Neuro Oncol.* 2021;23(8):1371–1382.

- Whittle IR, Smith C, Navoo P, Collie D. Meningiomas. Lancet (London, England). 2004;363(9420):1535–1543.
- Ali UM, Withrow DR, Judge AD, Plaha P, Darby SC. Temporal trends in the incidence of malignant and nonmalignant primary brain and central nervous system tumors by the method of diagnosis in England, 1993– 2017. *Neuro-Oncology*. 2023;25(6):1177–1192.
- Withrow DR, Devesa SS, Deapen D, et al. Nonmalignant meningioma and vestibular schwannoma incidence trends in the United States, 2004-2017. *Cancer.* 2021;127(19):3579–3590.
- Poon MTC, Brennan PM, Jin K, Sudlow CLM, Figueroa JD. Might changes in diagnostic practice explain increasing incidence of brain and central nervous system tumors? A population-based study in Wales (United Kingdom) and the United States. *Neuro Oncol.* 2021;23(6):979–989.
- Goldbrunner R, Minniti G, Preusser M, et al. EANO guidelines for the diagnosis and treatment of meningiomas. *Lancet Oncol.* 2016;17(9):e383–e391.
- Agarwal V, McCutcheon BA, Hughes JD, et al. Trends in management of intracranial meningiomas: analysis of 49,921 cases from modern cohort. *World Neurosurg*. 2017;106:145–151.
- Khan F, Amatya B. Factors associated with long-term functional outcomes, psychological sequelae and quality of life in persons after primary brain tumour. *J Neurooncol.* 2013;111(3):355–366.
- Sloane K, Vachani C, Hampshire MK, Metz JM, Hill-Kayser CE. Late effects in survivors of central nervous system tumors: reports by patients and proxies. *J Cancer Surviv*. 2016;10(2):234–240.
- Cantisano N, Menei P, Roualdes V, et al. Patient-reported functional executive challenges and caregiver confirmation in adult brain tumor survivors. *J Cancer Surviv*. 2021;15(5):696–705.
- Wiemann G, Pertz M, Kowalski T, et al. Complete response to therapy: why do primary central nervous system lymphoma patients not return to work? *J Neurooncol.* 2020;149(1):171–179.
- Gehrke AK, Baisley MC, Sonck AL, Wronski SL, Feuerstein M. Neurocognitive deficits following primary brain tumor treatment: systematic review of a decade of comparative studies. *J Neurooncol.* 2013;115:135–142.
- Benz LS, Wrensch MR, Schildkraut JM, et al. Quality of life after surgery for intracranial meningioma. *Cancer*. 2018;124(1):161–166.
- Konglund A, Rogne SG, Lund-Johansen M, et al. Outcome following surgery for intracranial meningiomas in the aging. *Acta Neurol Scand*. 2013;127(3):161–169.

- Zamanipoor Najafabadi AH, van der Meer PB, Boele FW, et al; Dutch Meningioma Consortium. Determinants and predictors for the long-term disease burden of intracranial meningioma patients. *J Neurooncol.* 2021;151(2):201–210.
- Nassiri F, Price B, Shehab A, et al; International Consortium on Meningiomas. Life after surgical resection of a meningioma: a prospective cross-sectional study evaluating health-related quality of life. *Neuro Oncol.* 2019;21(Suppl 1):i32–i43.
- van Alkemade H, de Leau M, Dieleman EM, et al. Impaired survival and long-term neurological problems in benign meningioma. *Neuro Oncol.* 2012;14(5):658–666.
- Waagemans ML, van Nieuwenhuizen D, Dijkstra M, et al. Long-term impact of cognitive deficits and epilepsy on quality of life in patients with low-grade meningiomas. *Neurosurgery*. 2011;69(1):72–8.
- 19. brainstrust. The story behind the data Report 1: Non-malignant brain tumours (NMBTs). 2018.
- Boone M, Roussel M, Chauffert B, Le Gars D, Godefroy O. Prevalence and profile of cognitive impairment in adult glioma: a sensitivity analysis. *J Neurooncol.* 2016;129:123–130.
- Bret P, Guyotat J, Madarassy G, Ricci A-C, Signorelli F. Tentorial meningiomas. Report on twenty-seven cases. *Acta Neurochir.* 2000;142:513–526.
- 22. Mariani M, Collins MWG. Neuropsychological profiles of breast cancer and brain tumor cohorts in Northeast Ontario, Canada. *Support Care Cancer.* 2018;26:3801–3809.
- 23. Wu A, Garcia MA, Magill ST, et al. Presenting symptoms and prognostic factors for symptomatic outcomes following resection of meningioma. *World Neurosurg.* 2018;111:e149–e159.
- Basalathullah MA, Malik M, Valiyaveettil D, Elizabeth NB, Ahmed SF. Return to work in survivors of primary brain tumours treated with intensity modulated radiotherapy. *Cancer Treat Res Commun.* 2021;26:100302.
- 25. Zamanipoor Najafabadi AH, van de Mortel JPM, Lobatto DJ, et al. Unmet needs and recommendations to improve meningioma care through patient, partner, and health care provider input: a mixed-method study. *Neurooncol. Pract.* 2020;7(2):239–248.
- Fry A, Littlejohns TJ, Sudlow C, et al. Comparison of sociodemographic and health-related characteristics of UK Biobank participants with those of the general population. *Am J Epidemiol.* 2017;186(9):1026–1034.

9

Advances